

**Comments To
Authors**

The question of the role of myeloid-derived cells which might be involved in suppression of cancer is extremely important. Current review represents an concise up-to-date review of current literature.

**Comments To
Authors**

A timely and well-written editorial article

Dear Editors,

We thank you for receiving so positively our editorial entitled "Myeloid Derived Suppressor Cells in Breast Cancer: A Novel Therapeutic Target?".

We have checked our writing and conclusions and occasionally improved the expression or clarity but not changed the content. Through bridging oncology and immunology, we are convinced that this contribution will be of genuine interest to a wide audience.

Respectfully yours,

